Author’s response to reviews

Title: Development of a framework for the co-production and prototyping of public health interventions

Authors:

Jemma Hawkins (hawkinsj10@cardiff.ac.uk)
Kim Madden (maddenk@cardiff.ac.uk)
Adam Fletcher (FletcherA@cardiff.ac.uk)
Luke Midgley (MidgleyLS@cardiff.ac.uk)
Aimee Grant (GrantA2@cardiff.ac.uk)
Gemma Cox (Gemma.Cox@wales.nhs.uk)
Laurence Moore (laurence.moore@glasgow.ac.uk)
Rona Campbell (Rona.Campbell@bristol.ac.uk)
Simon Murphy (muprhys7@cardiff.ac.uk)
Chris Bonell (chris.bonell@lshtm.ac.uk)
James White (whitej11@cardiff.ac.uk)

Version: 3 Date: 15 Aug 2017

Author’s response to reviews:

Dear Editor,

I am pleased to enclose the third revision of the manuscript “Development of a framework for the co-production and prototyping of public health interventions” (PUBH-D-16-03802.R1).

I have made the requested change in the manuscript as follows:
Editor Comments:

As two of the authors bear the same initials (LM), we ask that you define one of the authors with these initials within your 'Authors' Contributions' section so it is clear which of the two authors is being referred to.

The two authors with the same initials have been distinguished from each other in this section as requested and now appear as ‘LMidgley’ and ‘LMoore’ (Declarations: Authors’ Contributions, p26, lines 576-580).

The change has been made to the manuscript but is not highlighted or marked as tracked changes.

We look forward to hearing from you,

Jemma Hawkins

pp. the co-authors